Electronic Version v1.1 Stylesheet Version v1.1

SUBMISSION TYPE:	NEW ASSIGNMENT
NATURE OF CONVEYANCE:	MERGER
EFFECTIVE DATE:	10/13/2004

CONVEYING PARTY DATA

Name	Formerly	Execution Date	Entity Type
Advanced TelCom, Inc.		10/13/2004	CORPORATION:

RECEIVING PARTY DATA

Name:	Eschelon Telecom, Inc.
Trading As:	TA Eschelon Telecom, Inc.
Street Address:	730 Second Avenue South
Internal Address:	Suite 900
City:	Minneapolis
State/Country:	MINNESOTA
Postal Code:	55402
Entity Type:	CORPORATION:

PROPERTY NUMBERS Total: 1

Property Type	Number	Word Mark
Serial Number:	76256508	TFIRE

CORRESPONDENCE DATA

Fax Number: (612)436-6816

Correspondence will be sent via US Mail when the fax attempt is unsuccessful.

Phone: 612-436-1888

Email: gazeller@eschelon.com

Correspondent Name: Ginny A. Zeller

730 Second Avenue South Address Line 1:

Address Line 2: Suite 900

Minneapolis, MINNESOTA 55402 Address Line 4:

NAME OF SUBMITTER: J. Jeffery Oxley

> **TRADEMARK REEL: 003458 FRAME: 0317**

900066459

Signature:	/jjoxley/
Date:	01/10/2007
Total Attachments: 44	
source=TFire Logo#page1.tif	
source=ATI Stock Purchase Agreement#pa	ge1.tif
source=ATI Stock Purchase Agreement#pa	ge2.tif
source=ATI Stock Purchase Agreement#pa	ge3.tif
source=ATI Stock Purchase Agreement#pa	ge4.tif
source=ATI Stock Purchase Agreement#pa	ge5.tif
source=ATI Stock Purchase Agreement#pa	ge6.tif
source=ATI Stock Purchase Agreement#pa	ge7.tif
source=ATI Stock Purchase Agreement#pa	ge8.tif
source=ATI Stock Purchase Agreement#pa	ge9.tif
source=ATI Stock Purchase Agreement#pa	ge10.tif
source=ATI Stock Purchase Agreement#pa	ge11.tif
source=ATI Stock Purchase Agreement#pa	ge12.tif
source=ATI Stock Purchase Agreement#pa	ge13.tif
source=ATI Stock Purchase Agreement#pa	ge14.tif
source=ATI Stock Purchase Agreement#pa	ge15.tif
source=ATI Stock Purchase Agreement#pa	ge16.tif
source=ATI Stock Purchase Agreement#pa	ge17.tif
source=ATI Stock Purchase Agreement#pa	ge18.tif
source=ATI Stock Purchase Agreement#pa	ge19.tif
source=ATI Stock Purchase Agreement#pa	ge20.tif
source=ATI Stock Purchase Agreement#pa	ge21.tif
source=ATI Stock Purchase Agreement#pa	ge22.tif
source=ATI Stock Purchase Agreement#pa	
source=ATI Stock Purchase Agreement#pa	ge24.tif
source=ATI Stock Purchase Agreement#pa	ge25.tif
source=ATI Stock Purchase Agreement#pa	ge26.tif
source=ATI Stock Purchase Agreement#pa	ge27.tif
source=ATI Stock Purchase Agreement#pa	ge28.tif
source=ATI Stock Purchase Agreement#pa	·
source=ATI Stock Purchase Agreement#pa	
source=ATI Stock Purchase Agreement#pa	
source=ATI Stock Purchase Agreement#pa	
source=ATI Stock Purchase Agreement#pa	·
source=ATI Stock Purchase Agreement#pa	·
source=ATI Stock Purchase Agreement#pa	~
source=ATI Stock Purchase Agreement#pa	
source=ATI Stock Purchase Agreement#pa	-
source=ATI Stock Purchase Agreement#pa	ge38.tif

source=ATI Stock Purchase Agreement#page39.tif source=ATI Stock Purchase Agreement#page40.tif source=ATI Stock Purchase Agreement#page41.tif source=ATI Stock Purchase Agreement#page42.tif source=ATI Stock Purchase Agreement#page43.tif

TFire[™]

STOCK PURCHASE AGREEMENT

BY AND BETWEEN

ADVANCED TELCOM GROUP, INC.

AND

ESCHELON TELECOM, INC.

October 13, 2004

TABLE OF CONTENTS

ARTICLE I D	DEFINITIONS	1
1.1	Definitions	1
1.2	Construction	
ARTICLE II I	PURCHASE AND SALE OF THE SHARES	
2.1	Purchase and Sale	
2.2	Consideration	
2.3	Closing	
2.4	Deliveries by the Seller	9
2.5	Deliveries by the Purchaser	9
2.6	Retained Assets	9
2.7	Retained Liabilities	9
2.8	Minimum Net Working Capital	10
ARTICLE III	RELATED MATTERS	10
3.1	Use of Names	10
3.2	No Ongoing or Transition Services	10
	REPRESENTATIONS AND WARRANTIES OF THE SELLER	
4.1	Organization and Qualification.	
4.2	Authorization; Enforceability	11
4.3	Capital Stock	11
4.4	Ownership of the Shares	12
4.5	No Violation or Conflict	12
4.6	Financial Statements	12
4.7	Accounts Receivable	12
4.8	Contracts	
4.9	Customers	
4.10	Suppliers and Vendors	12
4.11	Real Property	
4.12	Assets	
4.13	Permits	13
4.14	Intellectual Property	13
4.15	Labor Matters	13
4.16	Employee Benefit Plans	13
4.17	No Other Agreements to Purchase	13
4.18	No Known Material Adverse Effect	14
4.19	Brokers	14

4.20	Tax Matters	14
ARTICLE V	REPRESENTATIONS AND WARRANTIES OF THE PURCHASER	14
5.1	Organization and Qualification	1.4
5.2	Authorization; Enforceability	14
5.3	No Violation or Conflict	14
5.4	Investigation	14
5.5	InvestigationBrokers	15
ARTICLE VI	COVENANTS	15
6.1	Performance	15
6.2	Regulatory and Other Authorizations; Notices and Consents; Etc	15
6.3	Conduct of ATI Business Prior to the Closing	16
6.4	Access	16
6.5	Notification	16
6.6	Exclusivity	17
6.7	Financing.	17
6.8	Certain Tax Matters	17
ARTICLE VI	I EMPLOYEE MATTERS	22
7.1	Employment	22
7.2	WARN	22
7.3	WARN	23
7.4	Excluded Employees	23
7.5	Past Service	24
	ATI Benefit Plan	
ARTICLE VI	II CONDITIONS PRECEDENT TO CLOSING; TERMINATION	24
8.1	Conditions Precedent to the Obligations of the Purchaser	24
8.2	Conditions Precedent to the Obligations of the Seller	2 1 25
8.3	Termination	
ARTICLE IX	INDEMNIFICATION	27
9.1		
9.2	Survival	27
9.3	Indemnification	27
9.3 9.4	Procedures	28
9.4 9.5	Calculation of Damages	29
	Assignment of Claims	30
9.6	Fraud; Intentional Misrepresentation	30
9.7	Sole Remedy	30
ARTICLE X	MISCELLANEOUS	30
10.1	Notices	30

10.2	Entire Agreement	31
10.3	Binding Effect	31
10.4	Assignment	31
10.5	Modifications and Amendments	31
10.6	Waivers and Consents	32
10.7	No Third Party Beneficiary	
10.8	Severability	32
10.9	Publicity	32
10.10	Governing Law; Jurisdiction	32
10.11	Force Majeure	33
10.12	Execution by Facsimile; Counterparts	
10.13	Headings and Captions	
10.14	Expenses	
10.15	Interpretation	34
10.16	Further Assurances	34

EXHIBITS AND SCHEDULES

GECC Guaranty
Retained Assets
Retained Liabilities
Financial Statements
Leased Real Property
Permits
Intellectual Property
Labor Matters
Employee Benefit Plans
Accrual of Vacation, Sick Time and Floating Holidays
The Seller's Consents and Approvals
EBITDA
The Purchaser's Consents and Approvals

STOCK PURCHASE AGREEMENT

- **THIS STOCK PURCHASE AGREEMENT** (the "Agreement") is made and entered into as of October 13, 2004 by and between Advanced TelCom Group, Inc., a Delaware corporation (the "Seller"), and Eschelon Telecom, Inc., a Delaware corporation (the "Purchaser").
- **WHEREAS**, the Seller owns all of the issued and outstanding shares of common stock, par value \$0.001 per share (the "*Shares*"), of Advanced TelCom, Inc., a Delaware corporation, ("*ATI*");
- **WHEREAS**, ATI owns all of the issued and outstanding shares of common stock, par value \$0.001 per share (the "*SCS Shares*"), of Shared Communications Systems, Inc., an Oregon corporation ("*SCS*" and, together with ATI, the "*Companies*");
- WHEREAS, the Seller desires to sell to the Purchaser, and the Purchaser desires to purchase from the Seller, the Shares on the terms and conditions set forth in this Agreement;
- **WHEREAS**, the Purchaser has delivered to the Seller an earnest money payment in the amount of \$1.5 million (the "*Earnest Money*");
- WHEREAS, simultaneously herewith, GE Business Productivity Solutions, Inc., a Georgia corporation ("GEBPS"), and the Purchaser have entered into an asset purchase agreement (the "Asset Purchase Agreement") pursuant to which the Purchaser will acquire substantially all of the assets of GEBPS; and
- **WHEREAS**, NTFC Capital Corporation, a Delaware corporation ("NTFC") owns 6,780,541 shares of the Purchaser's Series A Preferred Stock, par value \$0.01 per share (the "Eschelon Preferred Shares"); and
- WHEREAS, simultaneously herewith, NTFC and the Purchaser have entered into a redemption agreement (the "Eschelon Preferred Shares Redemption Agreement"), pursuant to which the Purchaser will purchase the Eschelon Preferred Shares from GECC (or one or more of its Affiliates).
- NOW, THEREFORE, in consideration of the premises and the mutual covenants, representations and warranties contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto agree as follows:

ARTICLE I

DEFINITIONS

1.1 <u>Definitions</u>. In addition to terms defined elsewhere in this Agreement, the following terms when used in this Agreement shall have the respective meanings set forth below:

- "2003 Audit" has the meaning set forth in Section 8.1(f).
- "2003 Financial Statements" has the meaning set forth in Section 4.6.
- "2004 Interim Financial Statements" has the meaning set forth in Section 4.6.
- "2004 Review" has the meaning set forth in Section 8.1(f).
- "Accounts Receivable" means all of the Companies' accounts and notes receivable from any Person.
- "Affiliate" means, with respect to a specified Person, any other Person which, directly or indirectly through one or more intermediaries, controls, is controlled by or is under common control with such Person, and without limiting the generality of the foregoing, includes, with respect to the specified Person: (a) any other Person which beneficially owns or holds 10% or more of the outstanding voting securities or other securities convertible into voting securities of such Person, (b) any other Person of which the specified Person beneficially owns or holds 10% or more of the outstanding voting securities or other securities convertible into voting securities, or (c) any director, officer or employee of such Person.
 - "Agreement" has the meaning set forth in the Preamble.
- "Asset Purchase Agreement" has the meaning set forth in the recitals to this Agreement.
 - "Assets" has the meaning set forth in Section 4.12(a).
 - "ATT" has the meaning set forth in the recitals to this Agreement.
- "ATI Contracts" means all material Customer Contracts, Real Property Leases and Supplier Contracts.
 - "ATI Employees" has the meaning set forth in Section 7.1.
- "Business" means the business of providing telecommunications products and services to small and mid-sized business customers primarily in California, Nevada, Oregon and Washington.
- "Business Day" means any day other than Saturday, Sunday or any other day on which banks are required or authorized to be closed in the city of New York, New York.
- "Cash" means all cash and cash equivalents (including marketable securities and short-term investments) on hand or in banks or other depositories.
 - "Claim" has the meaning set forth in Section 9.3(a).
 - "Closing" has the meaning set forth in Section 2.3.

- "Closing Date" has the meaning set forth in Section 2.3.
- "Code" means the Internal Revenue Code of 1986, as amended.
- "Companies" has the meaning set forth in the recitals to this Agreement.
- "Confidentiality Agreement" means the confidentiality agreement, dated as of August 6, 2004, by and between the Purchaser and GE Vendor Financial Services, an Affiliate of the Seller.
- "Contract" means any contract, plan, undertaking, understanding, agreement, license, lease, note, mortgage or other binding commitment, whether written or oral.
- "Court" means any court or arbitration tribunal of the United States, any domestic state, or any foreign country, and any political subdivision thereof.
- "Customer Contracts" means all written sales orders, master services agreements, customer contracts or other similar Contracts entered into by either Company with all customers of the Companies used solely in the operation of the Business.
 - "Damages" has the meaning set forth in Section 9.2(a).
- "Database" means all data and other information recorded, stored, transmitted and retrieved in electronic form.
 - "Earnest Money" has the meaning set forth in the recitals of this Agreement.
- "EBITDA" means, during the measuring period, consolidated net income (including gain from extraordinary items) of the Companies <u>less</u> (1) interest income and (2) gain on sale of assets, <u>plus</u> (1) provision for income taxes recorded during such measuring period or allocated or allocable to the Companies with respect to such measuring period, (2) interest expense, (3) depreciation and (4) amortization, in each case as determined in accordance with the past practice and procedures of GECC.
 - "Employee Plans" has the meaning set forth in Section 4.16.
- "Eschelon Preferred Shares" has the meaning set forth in the recitals of this Agreement.
- "Eschelon Preferred Shares Redemption Agreement" has the meaning set forth in the recitals of this Agreement.
 - "Excluded Employees" has the meaning set forth in Section 7.3.
 - "Financial Statements" has the meaning set forth in Section 4.6.
 - "GE" means General Electric Company, a New York corporation.
 - "GEBPS" has the meaning set forth in the recitals of this Agreement.

"GECC" means General Electric Capital Corporation, a Delaware corporation.

"GECC Guaranty" means the Guaranty of GECC, substantially in the form attached hereto as Exhibit A.

"Governmental Authority" means any governmental or legislative agency or authority (other than a Court) of the United States, any domestic state, or any foreign country, and any political subdivision or agency thereof, and includes any authority having governmental or quasi-governmental powers, including any administrative agency or commission.

"Indemnification Statement" has the meaning set forth in Section 6.8(d).

"Indemnified Party" has the meaning set forth in Section 9.3(a).

"Indemnifying Party" has the meaning set forth in Section 9.3(a).

"Intellectual Property" means, in each case used solely in the operation of the Business, all: (a) Patents; (b) trademarks, service marks, domain names and internet protocol addresses; (c) copyrights; (d) Software; (e) trade secrets and confidential, technical or business information; (f) Databases proprietary to any Seller; (g) copies and tangible embodiments of all the foregoing, in whatever form or medium; (h) all rights to obtain and rights to apply for patents, and to register trademarks and copyrights; (i) all rights under any licenses, registered user agreements, technology or materials, transfer agreements, and other agreements or instruments with respect to items in (a) to (h) above; (j) all goodwill associated therewith; (k) all applications and registrations therefor; and (l) all rights to sue and recover and retain damages and costs and attorneys' fees for present and past infringement of any rights appurtenant to the Intellectual Property.

"Law" means all laws, statutes, ordinances and Regulations of any Governmental Authority including all decisions of Courts having the effect of law in each such jurisdiction.

"Leased Real Property" means, in each case used solely in the operation of the Business, the real property leased by either Company as a tenant pursuant to the Real Property Leases in locations at which either Company has employees or switches, together with, to the extent leased by either Company, all buildings and other structures, facilities or improvements currently or hereafter located thereon, all fixtures, systems, equipment and items of personal property of either Company attached or appurtenant thereto and all easements, licenses, rights and appurtenances relating to the foregoing.

"Licensed Intellectual Property" means all Intellectual Property licensed or sublicensed by either Company from a third party.

"Liens" means any mortgage, pledge, security interest, attachment, encumbrance, lien (statutory or otherwise), option, conditional sale agreement, right of first refusal, first offer, termination, participation or purchase, or charge of any kind (including any agreement to give any of the foregoing), provided, however, that the term "Lien" shall

not include: (a) Liens for Taxes, assessments and charges any Governmental Authority due and being contested in good faith and diligently by appropriate proceedings; (b) servitudes, easements, restrictions, rights-of-way and other similar rights in real property or any interest therein, provided the same are not of such nature as to materially adversely affect the use of the property subject thereto; (c) Liens with respect to leasehold interests and licensed products; (d) Liens for Taxes either not due and payable or due but for which notice of assessments has not been given; (e) undetermined or inchoate Liens, charges and privileges incidental to current construction or current operations and statutory Liens, charges, adverse claims, security interests or encumbrances of any nature whatsoever claimed or held by any Governmental Authority which have not at the time been filed or registered against the title to the asset or served upon the Seller pursuant to Law or which relate to obligations not due or delinquent; (f) assignments of insurance provided to landlords (or their mortgagees) pursuant to the terms of any lease, and Liens or rights reserved in any lease for rent or for compliance with the terms of such lease; (g) security given in the ordinary course of the operation of the Business to any public utility, municipality or Government Authority in connection with the operation of the Business, other than security for borrowed money; (h) deposits or pledges made in connection with, or to secure payment of, workers' compensation, unemployment insurance, old age pension or other social security programs mandated under applicable Laws; (i) restrictions on transfer of securities imposed by applicable state and federal securities Laws; (j) purchase money liens incurred in the ordinary course consistent with past practices; and (k) Liens of carriers, warehousemen, mechanics, materialmen and repairmen incurred in the ordinary course of business consistent with past practices.

"Litigation" means any suit, action, arbitration, cause of action, claim, complaint, criminal prosecution, investigation, inquiry, demand letter, governmental or other administrative proceeding, whether at law or at equity, before or by any Court, Governmental Authority, arbitrator or other tribunal.

"Material Adverse Effect" means any circumstance, change in, or effect on the Business or the Assets that, individually or in the aggregate with any other circumstances, changes in, or effects on the Business, is materially adverse to the business, operations, assets, results of operations or financial condition of the Business or the Assets, taken as a whole, provided that the term "Material Adverse Effect" shall not include any circumstance, change in, or effect resulting from (a) changes in general economic conditions or the financial markets, (b) the announcement or performance of this Agreement and the transactions contemplated hereby, (c) general changes or developments in the industry in which either Company operates, (d) changes in any tax laws or regulations or applicable accounting regulations or principles, (e) changes caused by conditions or terms imposed by third parties in connection with obtaining any authorization, consent, order, approval, guarantee, bond or estoppel certificate or (f) changes in general legal, regulatory, political, economic or business conditions that generally affect the industry in which either Company conducts business or in which the Business operates.

"Net Working Capital" has the meaning set forth in Section 2.8.

- "Non-Assumable Claim" has the meaning set forth in Section 9.3(f).
- "NTFC" has the meaning set forth in the recitals to this Agreement.
- "Order" means any judgment, order, writ, injunction, ruling, stipulation, determination, award or decree of or by, or any settlement under the jurisdiction of, any Court or Governmental Authority.
- "Owned Intellectual Property" means all Intellectual Property in and to which either Company has, or has a right to hold, right, title and interest.
- "Patents" means all national (including the United States) and multinational statutory invention registrations, patents, patent registrations and patent applications, including all reissues, divisions, continuations, continuations-in-part, extensions and reexaminations, and all rights therein provided by multinational treaties or conventions and all improvements to the inventions disclosed in each such registration, patent or application.
- "Permits" means, to the extent transferable under applicable Law, all rights of the Companies under permits, authorizations, approvals, registrations and licenses issued by any Governmental Authority (and pending applications for the foregoing).
- "Person" means any natural person, corporation, limited liability company, unincorporated organization, partnership, association, joint stock company, joint venture, trust or any other entity.
 - "Potential Contributor" has the meaning set forth in Section 9.5.
 - "Pre-Closing Period" has the meaning set forth in Section 6.8(a).
 - "Pre-Closing Period Returns" has the meaning set forth in Section 6.8(d).
 - "Purchase Price" has the meaning set forth in Section 2.2.
 - "Purchaser" has the meaning set forth in the Preamble.
 - "Purchaser Tax Group" has the meaning set forth in Section 6.8(a).
 - "Purchaser Basket" has the meaning set forth in Section 9.2(b).
 - "Purchaser Flexible Benefit Plans" has the meaning set forth in Section 7.1(g).
 - "Real Property Leases" has the meaning set forth in Section 4.11(b).
 - "Regulation" means any rule or regulation of any Governmental Authority.
 - "Retained Assets" has the meaning set forth in Section 2.6.
 - "Retained Marks" has the meaning set forth in Section 2.6(c).

- "Retained Liabilities" has the meaning set forth in Section 2.7.
- "SCS" has the meaning set forth in the recitals to this Agreement.
- "SCS Shares" has the meaning set forth in the recitals to this Agreement.
- "Section 338(h)(10) Election" has the meaning set forth in Section 6.8(a).
- "Seller" has the meaning set forth in the Preamble.
- "Seller Basket" has the meaning set forth in Section 9.2(a).
- "Seller Flexible Benefit Plans" has the meaning set forth in Section 7.1(g).
- "Shares" has the meaning set forth in the recitals to this Agreement.
- "Software" means, to the extent owned or licensed by either Company and used solely in the operation of the Business, any and all (a) computer programs, (b) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (c) descriptions, flow-charts and other work product used to design, plan, organize and develop any of the foregoing, (d) the technology supporting any Internet site(s) operated by or on behalf of either Company and (e) all documentation, including user manuals and training materials, relating to any of the foregoing.
 - "Straddle Period" has the meaning set forth in Section 6.8(a).
 - "Straddle Period Returns" has the meaning set forth in Section 6.8(d).
- "Subsidiary" of a specified Person means any other Person in which such Person owns, directly or indirectly, more than 50% of the outstanding voting securities or other securities convertible into voting securities, or which may effectively be controlled, directly or indirectly, by such Person.
- "Supplier Contracts" means all Contracts for the provision of products or services, including interconnection agreements, entered into with either Company with any supplier of goods, services, materials or supplies used solely in the operation of the Business.
 - "Survival Date" has the meaning set forth in Section 9.1.
- "Tangible Personal Property" means each item or distinct group of furniture, fixtures, equipment, switches, tools, supplies, vehicles and other tangible personal property used solely in the operation of the Business and owned or leased by either Company, wherever located.
- "Tax" means any and all federal, state, local, or foreign taxes, fees, levies, duties, tariffs, imposts, and other charges of any kind (together with any and all interest,

penalties, additions to tax and additional amounts imposed with respect thereto) imposed by any Governmental Authority or other taxing authority, including: taxes or other charges on or with respect to income, franchises, windfall or other profits, gross receipts, property, sales, use, capital stock, payroll, employment, disability, social security, workers' compensation, unemployment compensation, or net worth; taxes or other charges in the nature of excise, withholding, ad valorem, stamp, transfer, value added, or gains taxes; license, registration and documentation fees; and customs' duties, tariffs, and similar charges, whether computed on a separate or consolidated, unitary or combined basis or in any other manner, whether disputed or not and including any obligation to indemnify or otherwise assume or succeed to the Tax liability of any other Person.

"Tax Claim" has the meaning set forth in Section 6.8(f).

"Tax Indemnified Party" has the meaning set forth in Section 6.8(f).

"Tax Indemnifying Party" has the meaning set forth in Section 6.8(f).

"Tax Returns" means returns, reports and information statements, including any schedule or attachment thereto, with respect to Taxes required to be filed with the IRS or any other Governmental Authority or other taxing authority or agency, domestic or foreign, including consolidated, combined and unitary tax returns.

"Third Party Claim" has the meaning set forth in Section 9.3(b).

"Unrelated Accounting Firm" has the meaning set forth in Section 6.8(a).

1.2 <u>Construction</u>. Unless the context requires otherwise: (a) the gender (or lack of gender) of all words used in this Agreement includes the masculine, feminine and neuter; (b) references to Articles and Sections refer to Articles and Sections of this Agreement; (c) "include", "includes", and "including" shall mean "including, without limitation"; (d) definitions used herein defined in the plural shall be deemed to include the singular as the context may require and definitions used herein defined in the singular shall be deemed to include the plural as the context may require; and (e) references to Schedules and Exhibits are to the Schedules and Exhibits attached to this Agreement, each of which is incorporated herein and made a part hereof for all purposes.

ARTICLE II

PURCHASE AND SALE OF THE SHARES

- 2.1 <u>Purchase and Sale</u>. Subject to the terms and conditions set forth in this Agreement, at the Closing, the Seller agrees to sell, transfer and deliver to Purchaser, and Purchaser agrees to purchase, acquire and accept from the Seller, the Shares.
- 2.2 <u>Consideration</u>. The aggregate consideration for the sale of the Shares to Purchaser shall be \$45,480,500 (the "*Purchase Price*"). At the Closing (as herein defined), the

Purchaser shall pay the Purchase Price (less the Earnest Money) to the Seller by wire transfer of immediately available funds.

- 2.3 <u>Closing</u>. Subject to the terms and conditions of this Agreement, the closing of the transactions contemplated by this Agreement (the "Closing") shall take place at the offices of Piper Rudnick LLP, 1251 Avenue of the Americas, 29th Floor, New York, New York, 10020, at 10:00 A.M. on December 31, 2004 or at such other place or time or on such other date as the Seller and the Purchaser may mutually agree upon in writing (the day on which the Closing takes place being the "*Closing Date*").
- 2.4 <u>Deliveries by the Seller</u>. At the Closing, the Seller shall deliver or cause to be delivered to the Purchaser (unless delivered previously) the following:
- (a) The stock certificates representing all of the Shares, duly endorsed in blank or accompanied by stock powers duly executed in blank; and
- (b) All documents, instruments and writings required to be delivered by the Seller at or prior to the Closing pursuant to this Agreement.
- 2.5 <u>Deliveries by the Purchaser</u>. At the Closing, the Purchaser shall deliver or cause to be delivered to the Seller (unless previously delivered) the following:
- (a) The Purchase Price (less the Earnest Money) in accordance with Section 2.2 of this Agreement; and
- (b) All documents, instruments and writings required to be delivered by the Purchaser at or prior to the Closing pursuant to this Agreement.
- 2.6 <u>Retained Assets</u>. Prior to the Closing, ATI shall distribute to the Seller the following assets of the Companies (the "*Retained Assets*") as part of a plan of complete liquidation of ATI within the meaning of Section 332 of the Code as provided in Section 338(h)(10) of the Code and the Treasury regulations promulgated thereunder:
 - (a) Cash;
 - (b) the assets listed on Schedule 2.6(b);
- (c) all tradenames, trademarks, identifying logos and service marks related to or employing any part or variation of the "GE" name, the name of any Subsidiary or Affiliate of GE (other than the Companies and/or pursuant to any other agreement between any Subsidiary or Affiliate of GE and the Purchaser or any Affiliate of the Purchaser) or any similar tradename, trademark, logo and service mark and any other tradenames, trademarks, identifying logos and service marks not used solely in connection with the Business (the "Retained Marks"); and
 - (d) any other assets, rights, or properties not solely related to the Business.
- 2.7 <u>Retained Liabilities</u>. Prior to the Closing, ATI shall transfer and/or assign to Seller the liabilities described on <u>Schedule 2.7</u> (the "*Retained Liabilities*").

2.8 <u>Minimum Net Working Capital</u>. Seller covenants that Net Working Capital on the Closing Date shall not be less than negative \$100,000. As used in this Section 2.8, "*Net Working Capital*" shall mean the sum of (A) the sum of (i) accounts receivable plus (ii) unbilled revenue <u>minus</u> (B) the sum of (i) accounts payable, (ii) accrued carrier costs, (iii) accrued customer billed and other taxes, (iv) accrued other and (v) current portion of capital leases. Each of the terms set forth in foregoing clauses (A) and (B) shall be derived from line items presented on the balance sheet of the Company to be prepared consistent with the books and records of the Company and delivered by Seller at Closing.

ARTICLE III

RELATED MATTERS

3.1 Use of Names.

- (a) It is expressly agreed that the Purchaser shall acquire pursuant to this Agreement all of the Companies' rights, title or interest in the tradenames, trademarks, identifying logos and service marks used solely in connection with the Business (excluding the Retained Marks). The Purchaser agrees that neither it nor any of its Affiliates shall make any use of the Retained Marks from and after the Closing Date. Simultaneously with the Closing, without limiting the foregoing, the Purchaser shall remove any reference to the Retained Marks from premises, letterhead, domain names and websites of the Companies.
- (b) The Purchaser acknowledges the Seller's (and their respective Affiliates') rights in and title to the Retained Marks and agrees that it shall not do or cause to be done any act that in any manner might infringe, or impair the validity, scope, or title in the Retained Marks.
- (c) In the event the Purchaser or any Affiliate of the Purchaser violates any of its obligations under this Section 3.1, the Seller may proceed against it in Law or in equity for such damages or other relief as a Court may deem appropriate. The Purchaser acknowledges that a violation of this Section 3.1 may cause the Seller irreparable harm that may not be adequately compensated for by money damages. The Purchaser hereby agrees that in the event of any actual or threatened violation of this Section 3.1, the Seller shall be entitled, in addition to other remedies that it may have, to a temporary restraining order and to a preliminary and final injunctive relief against the Purchaser or such Affiliate of the Purchaser to prevent any violations of this Section 3.1, without the necessity of posting a bond.
- 3.2 <u>No Ongoing or Transition Services</u>All data processing, accounting, insurance, banking, personnel, legal, communications and other services provided to the Companies by the Seller or any Affiliate of the Seller, including any agreements or understandings (written or oral) with respect thereto, shall terminate upon the Closing.

10

TE

ARTICLE IV

REPRESENTATIONS AND WARRANTIES OF THE SELLER

The Purchaser agrees and acknowledges that, except as expressly set forth in this Agreement, the Purchaser is acquiring the Companies and the Assets on an "as-is, where-is" basis. As an inducement to the Purchaser to enter into this Agreement and to consummate the transactions contemplated hereby, the Seller represents and warrants to the Purchaser as follows:

4.1 <u>Organization and Qualification.</u>

- (a) The Seller is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
- (b) ATI is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. SCS is a corporation duly organized, validly existing and in good standing under the laws of the state of Oregon.

4.2 <u>Authorization; Enforceability.</u>

- (a) The Seller has the corporate power and authority to own, hold, lease and operate its properties and assets and to carry on its business as currently conducted. The Seller has the corporate power and authority to execute, deliver and perform this Agreement. The execution, delivery and performance of this Agreement and the consummation of the transactions contemplated herein have been duly authorized and approved by the Seller and no other action on the part of the Seller is necessary in order to give effect thereto. This Agreement has been duly executed and delivered by, and constitutes the legal, valid and binding obligations of the Seller, enforceable against the Seller, in accordance with its terms, except as such enforcement may be limited by bankruptcy, insolvency or other similar laws affecting the enforcement of creditors' rights generally and except that the availability of equitable remedies is subject to the discretion of the court before which any proceeding therefor may be brought.
- (b) Each Company has the corporate power and authority to own, hold, lease and operate its properties and assets and to carry on the Business as currently conducted.

4.3 Capital Stock.

(a) There are (i) 3,000 Shares of ATI issued and outstanding, all of which are owned by the Seller, and (ii) 3,000 shares of SCS Shares issued and outstanding, all of which are owned by ATI. The Shares have been validly issued and are fully paid and non-assessable. There are no other shares of capital stock of ATI or SCS issued or outstanding. There are no shares of capital stock of ATI or SCS held as treasury shares. There are not any outstanding securities convertible into, exchangeable for, or carrying the right to acquire, equity securities of ATI or SCS, nor are there any subscriptions, warrants, options, rights or other arrangements or commitments which could obligate ATI or SCS to issue any shares of its capital stock or other equity interests.

- (b) Other than SCS, ATI does not own, directly or indirectly, any capital stock, membership interest or any other equity or debt securities of any corporation, firm, partnership, limited liability company, joint venture, association or other entity.
- 4.4 <u>Ownership of the Shares</u>. The Seller has good title to the Shares, and shall at the Closing own the Shares free and clear of all Liens.
- 4.5 No Violation or Conflict. None of the (a) execution and delivery of this Agreement by the Seller, (b) consummation of the transactions contemplated by this Agreement by the Seller and (c) performance by the Seller of its obligations under this Agreement, will (i) conflict with or violate the organizational documents of the Seller, ATI or SCS, as applicable, (ii) conflict with or violate any Law, Order or Permit applicable to the Seller, ATI or SCS or by which any Asset is bound or (iii) violate any order, injunction, decree, statute, rule or regulation applicable to the Seller, ATI or SCS, excluding from the foregoing clauses (ii) and (iii) such requirements, violations, conflicts, defaults or rights which would not individually or in the aggregate have a Material Adverse Effect.
- 4.6 <u>Financial Statements</u>. Attached hereto as <u>Schedule 4.6</u> are (a) the unaudited balance sheet and related statements of income of the Companies for the period from May 15, 2003 through December 31, 2003 (the "2003 Financial Statements"); and (b) the unaudited balance sheet and related statements of income of the Companies for the nine month period ended September 30, 2004 (the "2004 Interim Financial Statements" and, together with the 2003 Financial Statements, collectively, the "Financial Statements"). The Financial Statements have been prepared for internal management purposes from the books and records of the Companies.
- 4.7 <u>Accounts Receivable</u>. ATI has made available to the Purchaser a list of the Accounts Receivable as of September 20, 2004. All Accounts Receivable shall at the Closing be owned by the Companies free and clear of any Liens.
- 4.8 <u>Contracts</u>. ATI has made available to the Purchaser a list of all written contracts and agreements related solely to the operation of the Business, as of the date hereof, to which either Company is a party requiring payment or receipt by the Companies in excess of \$25,000 per annum.
- 4.9 <u>Customers</u>. ATI has made available to the Purchaser the names and addresses of substantially all of the customers of the Companies as of August 4, 2004. ATI has made available to the Purchaser substantially all Customer Contracts.
- 4.10 <u>Suppliers and Vendors</u>. ATI has made available the names and addresses of substantially all suppliers and vendors of the Companies, as of the date hereof, from which the Companies received or licensed supplies, goods, services in calendar year 2004. ATI has made available to the Purchaser substantially all Supplier Contracts.

4.11 Real Property.

(a) <u>Schedule 4.11(a)</u> lists the street address of each parcel of Leased Real Property.

(b) ATI has made available to the Purchaser copies of all real property leases to which either Company is a party (the "*Real Property Leases*") and all ancillary documents pertaining thereto in the possession of the Companies.

4.12 Assets.

- (a) Other than the Assets owned by ATG and used solely in the operation of the Business that will be contributed to the Companies at or prior to Closing, either ATI or SCS (i) owns, leases or has the legal right to use all of the assets (including Contracts) used solely in the operation of the Business (the "Assets") and (ii) with respect to Assets that are contractual rights, is a party to and enjoys the right to the benefits of all ATI Contracts.
- (b) Except as set forth on <u>Schedule 2.6(b)</u>, the Assets constitute all of the material properties, assets and rights used solely in the operation of the Business.
- (c) All of the Tangible Personal Property shall be owned by the Companies free and clear of any Liens as of the Closing.
- 4.13 <u>Permits</u>. <u>Schedule 4.13</u> lists, as of October 6, 2004, all material Permits and all pending applications therefor obtained by ATI and SCS that are used solely in the operation of the Business.
- 4.14 <u>Intellectual Property</u>. <u>Schedule 4.14</u> sets forth, as indicated therein, all of the material Owned Intellectual Property and the material Licensed Intellectual Property. Except as set forth on <u>Schedule 4.14</u> and other than the rights of software licensors or developers in software related to such Intellectual Property, either ATI or SCS owns or has the legal right to use all of its Intellectual Property and has a valid license to use all of its Licensed Intellectual Property.
- 4.15 <u>Labor Matters</u>. <u>Schedule 4.15</u> lists, as of the date hereof, by identifying code, place of employment, current annual salary, bonuses, accrued vacation, the date of employment of each employee of ATI and each GEBPS employee that will be transferred to ATI at or prior to the Closing.
- 4.16 Employee Benefit Plans. Schedule 4.16 lists all employee benefit plans (as defined in Section 3(3) of ERISA) and all bonus, stock or other security option, stock or other security purchase, stock or other security appreciation rights, incentive, deferred compensation, retirement or supplemental retirement, severance, golden parachute, vacation, cafeteria, dependent care, medical care, employee assistance program, education or tuition assistance programs, insurance and other similar fringe or employee benefit plans, programs or arrangements providing coverage for the benefit of any employee of ATI (each, an "Employee Plan," and together, the "Employee Plans"). ATI has made available to the Purchaser an accurate written summary of each Employee Plan.
- 4.17 <u>No Other Agreements to Purchase</u>. No Person, other than the Purchaser pursuant to this Agreement, has been granted by the Seller, ATI or SCS any written or oral agreement or option or any right or privilege capable of becoming an agreement or option for the purchase of the Shares or the purchase or acquisition of any of the Assets, other than orders for

goods or services accepted by ATI or SCS in the ordinary course of business, consistent with past practice.

- 4.18 <u>No Known Material Adverse Effect</u>. To the actual knowledge of any of Dave O'Neill, Eric Russell and Dan Wigger, there does not exist any event, fact or circumstance that has not been disclosed to the Purchaser and which has resulted in or could be reasonably expected to result in a Material Adverse Effect.
- 4.19 <u>Brokers</u>. Neither the Seller nor ATI has employed any financial advisor, broker or finder, and neither the Seller nor ATI has not incurred nor will incur any broker's, finder's, investment banking or similar fees, commissions or expenses in connection with the consummation of the transactions contemplated by this Agreement.
- 4.20 <u>Tax Matters</u>. For U.S. federal income tax purposes, ATI currently is, and, for the taxable year which includes the Closing Date, will be, a member of the affiliated group which includes the Seller, and such group will file a consolidated federal income tax return.

ARTICLE V

REPRESENTATIONS AND WARRANTIES OF THE PURCHASER

In order to induce the Seller to enter into this Agreement and to consummate the transactions contemplated hereby, the Purchaser represents and warrants to the Seller as follows:

- 5.1 <u>Organization and Qualification</u>. The Purchaser is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware.
- Authorization; Enforceability. The Purchaser has the corporate power and authority to execute, deliver and perform this Agreement. The execution, delivery and performance of this Agreement and the consummation of the transactions contemplated herein and therein have been duly authorized and approved by all necessary corporate or other action by the Purchaser (including, without limitation, the Purchaser's Board of Directors). This Agreement has been duly executed and delivered by, and constitute the legal, valid and binding obligations of, the Purchaser, enforceable against the Purchaser, in accordance with their terms, except as such enforcement may be limited by bankruptcy, insolvency or other similar laws affecting the enforcement of creditors' rights generally and except that the availability of equitable remedies is subject to the discretion of the court before which any proceeding therefor may be brought.
- 5.3 No Violation or Conflict. None of (a) the execution and delivery by the Purchaser of this Agreement; (b) consummation by the Purchaser of the transactions contemplated by this Agreement; or (c) the performance of this Agreement will (i) conflict with or violate the Certificate of Incorporation or Bylaws of the Purchaser, or (ii) conflict with , violate or require consent or approval pursuant to any Law, Order or Permit applicable to the Purchaser or any note, loan agreement, contract, or agreement to Purchaser is party or its properties or assets are bound.

- 5.4 <u>Investigation</u>. The Purchaser acknowledges and agrees that it (i) has made its own inquiry and investigation into, and, based thereon, has formed an independent judgment concerning the Companies, the Assets and the Business and (ii) has been furnished with or given adequate access to such information about the Companies, the Assets and the Business as it has requested. The Purchaser further acknowledges and agrees that the only representations, warranties, covenants and agreements made by the Seller are the representations, warranties, covenants, and agreements made in this Agreement and the Seller makes no express or implied representation or warranty with respect to the Companies, the Assets, the Business or otherwise or with respect to any other information provided by the Seller or its Affiliates including as to (a) the operation of the Business by the Purchaser after the Closing in any manner or (b) the probable success or profitability of the ownership, use or operation of the Business by the Purchaser after the Closing. The Purchaser has not relied upon any other representations or other information made or supplied by or on behalf of the Seller of any its Affiliates.
- 5.5 <u>Brokers</u>. The Purchaser has not employed any financial advisor, broker or finder, and the Purchaser has not incurred, and will not incur, any broker's, finder's, investment banking or similar fees, commissions or expenses in connection with the transactions contemplated by this Agreement.

ARTICLE VI

COVENANTS

6.1 <u>Performance</u>. Subject to the terms and conditions provided in this Agreement, each of the parties to this Agreement shall exercise reasonable commercial efforts to take or cause to be performed and fulfilled those of the conditions precedent to its obligations to consummate the transactions contemplated by this Agreement that are dependent upon its actions, including obtaining all necessary consents and approvals, to the end that the transactions contemplated hereby shall be fully and timely consummated.

6.2 Regulatory and Other Authorizations; Notices and Consents; Etc.

- (a) The Purchaser shall, and the Seller shall cause the Companies, to use their respective reasonable best efforts to obtain all authorizations, consents, orders and approvals of all Governmental Authorities and other Persons that may be or become necessary for its execution and delivery of, and the performance of its obligations pursuant to, this Agreement and shall cooperate fully with each other in promptly seeking to obtain all such authorizations, consents, orders and approvals. In addition, the Purchaser, at or prior to the Closing, shall use its reasonable best efforts to cause all guaranties, bonds (whether performance, financial or otherwise) and letters of credit with respect to the operations of ATI, SCS or the Business by any Affiliates of ATI (other than SCS) to be terminated and shall or shall cause an Affiliate of the Purchaser to guaranty such obligations and deliver such bonds and letters of credit as may be required with respect thereto.
- (b) The Purchaser shall, and the Seller shall cause ATI, to give promptly such notices to third parties and exercise commercially reasonable efforts to obtain such third party

consents and estoppel certificates as are reasonably necessary in connection with the consummation of the transactions contemplated by this Agreement.

6.3 Conduct of ATI Business Prior to the Closing. Between the date hereof and the Closing, except as expressly required or permitted by this Agreement or unless the Purchaser shall otherwise agree in writing, the Seller shall cause the Companies to conduct Business only in the ordinary course of business consistent with past practice and policies, including with respect to collection of Accounts Receivable and processing of accounts payables; provided, however, it is understood and agreed that, at or prior the Closing, (i) all Cash of the Companies shall be distributed to the Seller, (ii) certain GEBPS employees shall be transferred to ATI, (iii) certain Assets held by the Seller shall be transferred and/or contributed to ATI and (iv) intercompany debt shall be transferred or distributed to the Seller.

6.4 Access.

- (a) From the date hereof, during normal business hours and upon reasonable notice, the Seller shall cause ATI and each of ATI's officers, directors, employees, and agents to: (i) afford the Purchaser's officers and employees and authorized agents, accountants, counsel and Jefferies & Co. (solely in its capacity as a financing source for the transactions contemplated by this Agreement), to the extent they have agreed in writing to be bound by the terms of the Confidentiality Agreement, reasonable access, during normal business hours, to the offices, properties, plants, other facilities, books and records of ATI and SCS and to those officers, directors, employees, and agents of ATI and SCS who have any material knowledge relating to the Business; and (ii) furnish to such parties such information regarding ATI, SCS and/or the Business as the Purchaser and its agents may from time to time reasonably request, in the case of both clauses (i) and (ii), including for the purposes of completing audits.
- (b) For a period of six years after the Closing, the Purchaser shall retain the books and records of ATI and SCS that are transferred to the Purchaser pursuant to this Agreement relating to periods prior to the Closing in a manner reasonably consistent with the prior practices of ATI and (ii) upon reasonable notice, afford the officers, employees and authorized agents and representatives of ATI (or any of its Affiliates) reasonable access (including the right to make photocopies at ATI's expense), during normal business hours, to such books and records.
- 6.5 <u>Notification</u>. From the date hereof until the Closing, the Purchaser shall notify and the Seller shall cause ATI to notify the other parties to this Agreement in writing of the occurrence, or pending or threatened occurrence, of (a) any event that would constitute a breach or violation of this Agreement by any party or that could reasonably be anticipated to cause any representation or warranty made by the notifying party in this Agreement to be false or misleading in any respect (including any event or circumstance which would have been required to be disclosed if such event or circumstance occurred or existed on or prior to the date of this Agreement); and (b) all other material developments affecting the Assets or the Business. Any such notification shall not limit or alter any of the representations, warranties or covenants of the parties set forth in this Agreement nor any rights or remedies a party may have with respect to a breach of any representation, warranty or covenant.

- 6.6 Exclusivity. Subject to and in consideration of receipt of the Earnest Money, the Seller shall not, and shall cause ATI and SCS not to, directly or indirectly, through any officer, director, manager or agent of any of them or otherwise, initiate, solicit or encourage (including by way of furnishing non-public information or assistance), or enter into negotiations of any type, directly or indirectly, or enter into a confidentiality agreement, letter of intent or purchase agreement, merger agreement or other similar agreement with any Person other than the Purchaser with respect to a sale of all or any substantial portion of the Assets, or a merger, consolidation, business combination, sale of all or any substantial portion of the capital stock of ATI, or the liquidation or similar extraordinary transaction with respect to ATI.
- 6.7 <u>Financing</u>. The Purchaser shall use its reasonable best efforts to obtain the financing described in Section 8.1(e). The Seller shall assist the Purchaser with all reasonable requests with respect to the Purchaser's effort to obtain financing.

6.8 Certain Tax Matters

- (a) <u>Certain Definitions</u>. As used in this Agreement:
- (i) "Purchaser Tax Group" means the affiliated group, within the meaning of Section 1504(a) of the Code, of which the Purchaser is the common parent;
- (ii) "Pre-Closing Period" means any taxable period, including that portion of any Straddle Period, which ends on or before the Closing Date;
- (iii) "Section 338(h)(10) Election" means the election to be made by the Purchaser and the Seller pursuant to Section 338(h)(10) of the Code, and any corresponding provision of state or local tax law, as described in Section 6.8(b) of this Agreement; and
- (iv) "Straddle Period" means any taxable period that includes (but does not end on) the Closing Date.
 - (v) "Unrelated Accounting Firm" means Grant Thornton LLP.
- (b) Section 338(h)(10) Election. The Seller shall join with the Purchaser in making the Section 338(h)(10) Election to treat the transaction hereunder as the deemed sale of the assets of ATI for U.S. federal and, to the extent applicable, state income tax purposes. The Seller and the Purchaser agree (i) to execute, at the Closing or as promptly thereafter as practicable, all forms of any nature necessary to effectuate such elections (including, without limitation, Internal Revenue Service Form 8023 and any comparable Forms under applicable state or local law), and (ii) to cooperate with each other and take all other actions necessary and appropriate (including filing such Forms and any other documents as may be required) to effect and preserve a timely Section 338(h)(10) Election in accordance with applicable law. The Purchaser and the Seller shall each cause such Forms or other documents to be duly executed by an authorized person. The Purchaser and the Seller shall report the acquisition of the Shares pursuant to the Agreement consistent with the Section 338(h)(10) Election, and shall take no position inconsistent with such treatment.

(c) <u>Allocation of Purchase Price</u>. The parties shall agree to the allocation of the Purchase Price among the assets of ATI at or prior to Closing, and shall report the transaction in a manner consistent with such allocation.

(d) <u>Return Filing, Refunds, Credits and Transfer Taxes.</u>

- (i) Except with regard to Tax Returns for Straddle Periods, the Seller shall prepare, or cause to be prepared, and file, or cause to be filed, on a timely basis all Tax Returns of ATI and SCS for all Pre-Closing Periods (the "Pre-Closing Period Returns"). The Seller shall pay, or cause to be paid, all Taxes with respect to ATI and SCS shown to be due on the Pre-Closing Period Returns.
- (ii) The Purchaser shall prepare, or cause to be prepared, and shall file, or cause to be filed, on a timely basis all Tax Returns other than the Pre-Closing Period Returns with respect to ATI, including Tax Returns, if any, for the Straddle Period (the "Straddle Period Returns"); provided, that any Straddle Period Return shall be prepared in a manner consistent with past practice, except as otherwise required by applicable Law. The Purchaser shall pay, or cause to be paid, all Taxes shown to be due on such Tax Returns.
- (iii) The Purchaser shall provide the Seller with copies of any Straddle Period Returns at least 45 business days prior to the due date thereof (giving effect to any extensions thereto), accompanied by a statement calculating in reasonable detail the Seller' indemnification obligation pursuant to Section 6.8(f) (the "Indemnification Statement"). The Seller shall have the right to review such Straddle Period Returns and Indemnification Statement prior to the filing of such Straddle Period Returns. If the Seller disputes any amounts shown to be due on such Tax Returns or the amount calculated in the Indemnification Statement, the Seller and the Purchaser shall consult and resolve in good faith any issues arising as a result of the review of such Straddle Period Return and Indemnification Statement. If the Seller agrees to the Indemnification Statement amount, the Seller shall pay to the Purchaser an amount equal to the Taxes shown on the Indemnification Statement (less any estimated tax payments made by the Seller or ATI in respect of such Taxes on or before the Closing Date) not later than three business days before the due date (including any extensions thereof) for payment of Taxes with respect to such Straddle Period Return. If the parties are unable to resolve any dispute within thirty (30) business days after the Seller's receipt of such Straddle Period Return and Indemnification Statement, such dispute shall be resolved by the Unrelated Accounting Firm, which shall resolve any issue in dispute as promptly as practicable. If the Unrelated Accounting Firm is unable to make a determination with respect to any disputed issue prior to the due date (including any extensions) for the filing of the Straddle Period Return in question, (A) the Purchaser shall file, or shall cause to be filed, such Straddle Period Return without such determination having been made and (B) the Seller shall pay to the Purchaser, not later than three days before the due date (including any extensions thereof) for the payment of Taxes with respect to such Straddle Period Return, an amount determined by the Seller as the proper amount chargeable to the Seller pursuant to this Section 6.8. Upon delivery to the Seller and the Purchaser by the Unrelated Accounting Firm of its determination, appropriate adjustments shall be

18

made to the amount paid by the Seller in accordance with the immediately preceding sentence in order to reflect the decision of the Unrelated Accounting Firm. The determination by the Unrelated Accounting Firm shall be final, conclusive and binding on the parties.

- (iv) The Seller and the Purchaser shall reasonably cooperate, and shall cause their respective Affiliates, officers, employees, agents, auditors and representatives reasonably to cooperate, in preparing and filing all Tax Returns (including amended returns and claims for refund), including maintaining and making available to each other all necessary records in connection with determining Taxes and in resolving all disputes and audits with respect to all taxable periods relating to Taxes. The Purchaser recognizes that the Seller will need access, from time to time, after the Closing Date, to certain accounting and tax records and information held by ATI to the extent such records and information pertain to events occurring prior to the Closing Date; therefore, without limiting the provisions set forth in Section 6.4, the Purchaser agrees that from and after the Closing Date the Purchaser shall, and shall cause ATI to, (A) retain and maintain such records until such time as the Seller reasonably determines that such retention and maintenance is no longer necessary and (B) allow the Seller and their agents and representatives (and agents and representatives of their Affiliates) reasonable access to inspect, review and make copies of such records as the Seller may deem necessary or appropriate from time to time. The Purchaser shall indemnify the Seller from and against any penalties, additions to tax or interest imposed on the Seller as a result of any failure of the Purchaser to provide tax records or other information to the Seller in a timely manner.
- (v) The Purchaser shall not, and shall cause ATI not to, dispose of or destroy any of the business records and files of ATI relating to Taxes in existence on the Closing Date without first offering to turn over possession thereof to the Seller by written notice to the Seller at least 30 days prior to the proposed date of such disposition or destruction.
- (vi) Any refunds and credits of Taxes of ATI or similar benefit (including any interest or similar benefit) received from or credited thereon by the applicable tax authority with respect to (A) any taxable period ending on or before the Closing Date or (B) Taxes for which the Seller have indemnified the Purchaser under the Agreement, shall be for the account of the Seller, and if received or utilized by the Purchaser or ATI, shall be paid to the Seller within five business days after the Purchaser or ATI receives such refund or utilizes such credit. Except as provided in the next sentence, any refunds or credits of ATI with respect to any Straddle Period shall be apportioned between the Seller, on the one hand, and the Purchaser, on the other hand, on the basis of an interim closing of the books. In the case of a refund or credit attributable to any Taxes that are imposed on a periodic basis and are attributable to the Straddle Period, other than Taxes based upon or related to gross or net income or receipts, the refund or credit of such Taxes of ATI for the Pre-Closing Period shall be deemed to be the amount of such refund or credit for the Straddle Period multiplied by a fraction the numerator of which is the number of days in the Straddle Period ending on the Closing Date and the denominator of which is the number of days in the Straddle Period.

19

- (vii) Notwithstanding any other provisions of this Agreement to the contrary, all sales, use, transfer, gains, stamp, duties, recording and similar Taxes incurred in connection with the transactions contemplated by this Agreement (other than the transfers contemplated by Section 2.6) shall be borne equally by the Purchaser and ATI, and any Returns that must be filed in connection with such Taxes shall be prepared and filed when due by the party primarily or customarily responsible under applicable Law for filing such Returns, and such party shall provide such Returns to the other party for review at least ten days prior to the due date for such Returns and shall file those Returns only with the consent of the other party, which consent shall not be unreasonably withheld.
- (e) <u>Elections</u>. Except with respect to the Section 338(h)(10) Election, the Purchaser shall not, and shall cause ATI not to, make, amend, or revoke any Tax election if such action would adversely affect the Seller, or any Person (other than ATI) as to whom or with whom the Seller have filed a consolidated return, with respect to any taxable period ending on or before the Closing Date or for the Pre-Closing Period or any Tax refund with respect thereto.

(f) Tax Indemnification.

- (i) The Purchaser shall indemnify, defend and hold harmless the Seller and its Affiliates, at any time after the Closing, from and against any liability for Taxes of ATI and SCS for any taxable period ending after the Closing Date, including any Taxes resulting solely from a transaction undertaken by the Purchaser or its Affiliates with respect to ATI or SCS after the Closing that are not otherwise contemplated by this Agreement, except for Straddle Periods, in which case the Purchaser's indemnity will cover only that portion of any such Taxes that is not attributable to the Pre-Closing Period.
- (ii) The Seller shall indemnify, defend and hold harmless the Purchaser and its Affiliates, at any time after the Closing, from and against any liability for (A) Taxes of ATI and SCS, except as provided in Section 6.8(d)(vii) of the Agreement, for the Pre-Closing Period (including the portion of any Straddle Period ending on the Closing Date), including any Damages resulting from the distribution of the Retained Assets to the Seller pursuant to Section 2.6 hereof; and (B) Damages attributable to (1) any breach of the representation contained in Section 4.20 or (2) the failure of the Seller (or any member of the affiliated group that includes the Seller) to comply with Seller's obligations in Section 6.8(b).
- (iii) In determining the responsibility of the Seller and the Purchaser for Taxes attributable to any Straddle Period, Taxes based upon or related to gross or net income or receipts shall be apportioned on the basis of an interim closing of the books as of the Closing Date, and all other Taxes attributable to any Straddle Period shall be prorated on a daily basis.
- (iv) If a claim for Taxes shall be made by any taxing authority in writing, which, if successful, might result in an indemnity payment pursuant to this Section 6.8, the party seeking indemnification (the "Tax Indemnified Party") shall

promptly notify the other party (the "Tax Indemnifying Party") in writing of such claim (a "Tax Claim") within a reasonably sufficient period of time to allow the Tax Indemnifying Party effectively to contest such Tax Claim, and in reasonable detail to apprise the Tax Indemnifying Party of the nature of the Tax Claim, and provide copies of all correspondence and documents received by it from the relevant taxing authority. Failure to give prompt notice of a Tax Claim hereunder shall affect the Tax Indemnifying Party's obligation under this Section to the extent that the Tax Indemnifying Party is prejudiced by such failure to give prompt notice.

- With respect to any Tax Claim which might result in an indemnity payment to the Purchaser pursuant to this Section 6.8(f), the party potentially liable for such indemnification shall control all proceedings taken in connection with such Tax Claim and, without limiting the foregoing, may in its sole discretion and at its sole expense pursue or forego any and all administrative appeals, proceedings, hearings and conferences with any taxing authority with respect thereto, and may, in its sole discretion, either pay the Tax claimed and sue for a refund where applicable law permits such refund suits or contest such Tax Claim. The Tax Indemnified Party shall not under any circumstances settle or otherwise compromise any Tax Claim referred to in the preceding sentence without the Tax Indemnifying Party's prior written consent. In connection with any proceeding taken in connection with such Tax Claim, each party shall keep the other informed of all material developments and events relating to such Tax Claim if involving a material liability for Taxes. Each party shall cooperate with the other in contesting such Tax Claim, which cooperation shall include, without limitation, the retention and the provision of records and information which are reasonably relevant to such Tax Claim, and making employees available to provide additional information or explanation of any material provided hereunder or to testify at proceedings relating to such Tax Claim, provided that no charges shall be incurred for the services of such employees.
- (vi) The Seller and the Purchaser shall jointly control, at each party's own cost and expense, all proceedings taken in connection with any Tax Claim relating solely to Taxes attributable to a Straddle Period, and neither the Seller nor the Purchaser shall settle a Tax Claim relating to such Straddle Period Taxes without the prior written consent of the other party, which consent shall not be unreasonably withheld.
- (vii) The amount of any Taxes payable under this Section 6.8 shall be net of any tax benefit actually realized therefrom.
- (viii) All matters relating in any manner to Tax indemnification obligations and payment of Taxes shall be governed exclusively by this Section 6.8; provided, however, that any indemnity relating to Section 6.8(f)(ii)(B) shall be subject to the limitations on liability set forth in Section 9.2.
- (ix) The representations, warranties, covenants and agreements contained in this Section 6.8 shall survive until the expiration of the applicable statute of limitations (including extensions).

ARTICLE VII

EMPLOYEE MATTERS

- 7.1 <u>Employment.</u> All employees of ATI including the employees of GEBPS transferred to ATI at or prior to Closing ("ATI Employees") (other than Excluded Employees) shall be employed by the Purchaser or one of its Affiliates without any relocation, with positions and working conditions that are substantially comparable to their positions with ATI on terms and conditions to be established by the Purchaser, which shall include the following employee benefit and compensation arrangements:
- (a) <u>Compensation</u>. For each ATI Employee, commencing on the Closing Date and for a period of 12 months thereafter, provided that such ATI Employee continues in the employment of the Purchaser or one of its Affiliates for such period of time, the Purchaser shall provide to such ATI Employee (i) salary no less than 103% of the salary such ATI Employee is being paid as of the Closing Date and (ii) bonus and incentive compensation opportunities no less favorable than those provided to similarly situated employees of the Purchaser.
- (b) Benefit Plans. For each ATI Employee, commencing on the Closing Date and for a period of 12 months thereafter, provided that such ATI Employee continues in the employment of the Purchaser or one of its Affiliates for such period of time, the Purchaser shall provide to each ATI Employee benefit plans, including any associated employee coverages and benefit costs (such as premiums, out-of-pocket expenses and deductibles), that are no less favorable than those provided to similarly situated employees of the Purchaser.
- (c) <u>Health Plans</u>. Commencing on the Closing Date and for a period of twelve 12 months thereafter, provided that ATI Employees continue in the employment of the Purchaser or one of its Affiliates for such period of time, the Purchaser shall provide ATI Employees and their eligible dependents with coverage under group health plans and the Purchaser agrees that its or one of its affiliates' group health plan covering ATI Employees and their eligible dependents shall not contain any pre-existing condition limitation or exclusion applicable to their participation therein and shall give each ATI Employee credit toward applicable deductibles and annual out-of-pocket limits for expenses incurred in the calendar year in which the Closing Date occurs.
- (d) <u>Defined Contribution Plan Loans</u>. On or as soon as reasonably practicable after the Closing Date, the Purchaser or one of its Affiliates shall permit ATI Employees who have outstanding loans under the 401(k) plan provided by ATI to roll over their account balance (including such loans) to a 401(k) plan sponsored by the Purchaser.
- (e) <u>Severance and Lay-Off.</u> Commencing on the Closing Date, the Purchaser may terminate the employment of or lay off any ATI Employee, provided that, for a period commencing on the Closing Date and ending on the second anniversary thereof, any such ATI Employee terminated or laid off without cause shall be entitled to receive severance payments and severance health benefits no less favorable than, and on the same terms and conditions of, those being provided by ATI immediately prior to Closing.

- (f) <u>Vacation, Sick Time and Floating Holidays</u>. The Purchaser agrees that, for a period of at least 12 months following the Closing and for every succeeding year, provided such ATI Employee remains an employee of the Purchaser or one of its Affiliates, the vacation entitlement provided to ATI Employees shall be equal to the greater of the ATI Employee's current vacation entitlement under ATI's vacation policy in effect as of the Closing Date and the Purchaser's then current vacation policy. The Purchaser agrees that each ATI Employee's remaining unused annual vacation time, sick days and floating holidays will have accrued (earned but unused) in full as of the Closing Date and such accruals are set forth on <u>Schedule 7.1(f)</u> and shall be paid for by the Purchaser following the Closing in accordance with ATI's vacation, sick time and floating holiday policies.
- Closing Date, the Seller shall transfer, or cause to be transferred, to the Purchaser an amount, in cash, representing the aggregate 2004 contributions of each ATI Employee then participating in the GE flexible spending accounts (the "Seller Flexible Benefits Plan"), net of reimbursements. The Purchaser shall cause such amounts to be credited to each such employee's accounts under the Purchaser's (or one of its Affiliate's) corresponding health care and dependent care spending account plan (the "Purchaser Flexible Benefits Plan") which shall be established and in effect for such employees as of the Closing Date, and all claims for reimbursement which have not been paid as of the date of the transfer to the Purchaser and credited under the Purchaser Flexible Benefits Plan shall be paid pursuant to and under the terms of the Purchaser Flexible Benefits Plan. In connection with such transfer, Purchaser shall deem that such employees' deferral elections made under the Seller Flexible Benefits Plan for the 2004 calendar year shall continue in effect under the Purchaser Flexible Benefits Plan for the remainder of the 2004 calendar year following the Closing Date.
- (h) <u>Tuition Reimbursement</u>. Purchaser shall provide, or cause an Affiliate to provide, tuition reimbursement under the Purchaser's tuition reimbursement plan to ATI Employees for approved courses that commenced prior to the Closing Date and end following the Closing Date in accordance with the provisions of such plan; *provided, however*, that the reimbursement for such approved courses under the Purchaser's Reimbursement Plan shall be no less than what would have been if the reimbursement was provided under the GE tuition reimbursement plan. Purchaser shall provide, or cause an Affiliate to provide, tuition reimbursement under the Purchaser's Tuition Refund Plan to ATI Employees for approved courses commencing on or after the Closing Date in accordance with the provisions of such plan.
- 7.2 <u>WARN</u>. The Purchaser and its Affiliates shall take no action within six months following the Closing Date which, together with actions taken prior to the Closing Date by ATI, would result in any obligation or liability to the Seller or constitute a "plant closing" or a "mass layoff" under the Worker Adjustment and Retraining Notification Act, 29 U.S.C. § 2101 et seq. or any other similar federal, state or local statute, regulation or ordinance.
- 7.3 Excluded Employees. On or prior to the Closing Date, the Seller shall, or shall cause one of its Affiliates (other than ATI) to, hire any employee of ATI who is on short-term or long-term disability leave or military leave as of the Closing Date (the "Excluded *Employees*"). After the Closing, the Purchaser or one of its Affiliates shall offer employment to any Excluded Employee subject to the following conditions: (a) if on medical or disability leave,

such individual is released by his or her physician to return to active employment and (b) such individual actually reports for active employment with the Purchaser or one of its Affiliates promptly following such medical release or expiration of approved leave; provided, however, that the Purchaser and its Affiliates shall not be required to offer employment under this provision to any Excluded Employee who does not apply for such employment prior to the later of (i) twenty-four months after the Closing Date and (ii) the expiration of any period under applicable Law. The Seller shall retain liability and responsibility for any Excluded Employee until such employee becomes an employee of the Purchaser or one of its Affiliates or, if earlier, until such employee's employment with the Seller or its Affiliates, such Excluded Employee shall be deemed an ATI Employee, and Purchaser or one of its Affiliates shall assume all liabilities and responsibilities for such employee accruing on and after the date such Excluded Employee becomes an ATI Employee.

- 7.4 Past Service. If ATI Employees are included in any benefit plan or arrangement of Purchaser, the ATI Employees shall receive credit for service to ATI or any of its Affiliates prior to the effective time to the same extent such service was counted under similar plans for purposes of determining eligibility to participate, vesting and benefit accruals and entitlements, other than benefit accruals under any defined benefit pension plan (including service with a prior employer recognized by ATI or any of its Affiliates).
- 7.5 <u>ATI Benefit Plan</u>. Except as expressly provided in this Agreement, the Purchaser assumes no liability with respect to, and receives no right or interest in, any Employee Plan or any other "employee benefit plan" (as defined in Section 3(3) of ERISA) maintained or sponsored by ATI or any of its Affiliates. As of the Closing Date, all ATI Employees shall cease active participation in all Employee Plans.

ARTICLE VIII

CONDITIONS PRECEDENT TO CLOSING; TERMINATION

- 8.1 <u>Conditions Precedent to the Obligations of the Purchaser</u>. The obligation of the Purchaser to consummate the transactions described in this Agreement and any and all liability of the Purchaser to the Seller shall be subject to the fulfillment on or before the Closing of the following conditions precedent, each of which may be waived by the Purchaser in its sole discretion:
- (a) Representations, Warranties and Covenants. The representations and warranties contained in Article IV shall have been true and correct in all material respects when made and as of the Closing, with the same force and effect as if made as of the Closing Date, other than such representations and warranties that are expressly made as of another date which shall be true and correct in all material respects as of such date, and the material covenants and agreements contained in this Agreement to be complied with by the Seller at or before Closing shall have been complied with in all material respects, and the Purchaser shall have received a certificate from the Seller to such effect signed by a duly authorized officer of the Seller.

- (b) <u>No Material Adverse Effect</u>. To the actual knowledge of any of Dave O'Neill, Eric Russell and Dan Wigger, there shall not exist any event, fact or circumstance that has resulted in or could be reasonably expected to result in a Material Adverse Effect.
- (c) <u>No Actions, Suits or Proceedings</u>. No Order of any Court or Governmental Authority shall have been issued restraining, prohibiting, restricting or delaying, the consummation of the transactions contemplated by this Agreement. No Litigation shall be pending or before any Court or Governmental Authority to restrain, prohibit, restrict or delay, or to obtain damages or a discovery order in respect of this Agreement or the consummation of the transactions contemplated hereby.
- (d) <u>Closing Documents</u>. The Seller shall have delivered or caused to be delivered to the Purchaser the following:
 - (i) the resignations of all officers and members of the Boards of Directors of ATI;
 - (ii) the GECC Guaranty, duly executed by GECC;
 - (iii) a list, as of the Closing Date, by identifying code, place of employment, current annual salary, bonuses, accrued vacation, the date of employment of each employee of ATI;
 - (iv) a copy of the resolutions duly and validly adopted by the Board of Directors of the Seller, certified by its Secretary, authorizing and approving the execution, delivery and performance of this Agreement and the transactions contemplated hereby.
- (e) <u>Capital Raise</u>. The Purchaser shall have consummated a debt offering and/or equity transaction, or shall have secured other financing, resulting in net proceeds to the Purchaser of not less than \$57,000,000.
- Financial Statements of ATI for the period from May 15, 2003 through December 31, 2003, together with an unqualified audit opinion of KPMG LLP with respect to such Financial Statements (the "2003 Audit"), and (ii) ATI's Financial Statements for the nine-month period ended on September 30, 2004, together with (A) a review or similar non-attestation opinion of KPMG LLP with respect to such Financial Statements (the "2004 Review") and (B) audited Financial Statements of ATI for the ninth-month period ended on September 30, 2004, together with an unqualified audit opinion of KPMG LLP with respect to such Financial Statements. EBITDA determined based on the 2003 Audit and the 2004 Review, in each case as calculated on a consistent basis and as set forth on Schedule 8.1(f), (A) presented in the 2003 Audit shall not contain a negative variation of greater than \$1,000,000 from EBITDA presented in the 2003 Financial Statements and (B) presented in the 2004 Review shall not contain a negative variation of greater than \$750,000 from EBITDA presented in the 2004 Interim Financial Statements.
- 8.2 <u>Conditions Precedent to the Obligations of the Seller</u>. The obligations of the Seller to consummate the transactions described in this Agreement and any and all liability of the

Seller to the Purchaser shall be subject to the fulfillment on or before the Closing Date of the following conditions precedent, each of which may be waived by the Seller in its sole discretion:

- (a) Representations, Warranties and Covenants. The representations and warranties of the Purchaser contained in this Agreement shall have been true and correct in all material respects when made and as of the Closing, with the same force and effect as if made as of the Closing Date, other than such representations and warranties that are expressly made as of another date which shall be true and correct in all material respects as of such date, and the material covenants and agreements contained in this Agreement to be complied with by the Purchaser on or before the Closing shall have been complied with in all material respects, and the Seller shall have received a certificate from the Purchaser to such effect signed by a duly authorized officer thereof.
- (b) <u>Authorizations, Consents Orders, Approvals, Guaranties, Bonds and Estoppel Certificates</u>. Each of the authorizations, consents, orders, approvals, guaranties, bonds, letters of credit and estoppel certificates described on <u>Schedule 8.2(b)</u> shall have been obtained.
- (c) <u>No Actions, Suits or Proceedings</u>. No Order of any Court or Governmental Authority shall have been issued restraining, prohibiting, restricting or delaying, the consummation of the transactions contemplated by this Agreement. No Litigation shall be pending before any Court or Governmental Authority to restrain, prohibit, restrict or delay, or to obtain damages or a discovery order in respect of this Agreement or the consummation of the transactions contemplated hereby.
- (d) <u>Purchase Price</u>. The Purchaser shall have delivered the Purchase Price (less the Earnest Money) to the Seller as provided in Section 2.2.
- (e) <u>Cash Payment for Eschelon Preferred Shares</u>. The Purchaser shall have delivered the purchase price under the Eschelon Preferred Shares Redemption Agreement in accordance with the terms thereof and the closing contemplated therein shall have occurred.
- (f) <u>Management Services Agreement</u>. If the closing under the Asset Purchase Agreement shall not occur on the same date as the Closing, the Purchaser shall have delivered a duly executed Management Services Agreement (as defined in the Asset Purchase Agreement) to the Seller.
- (g) <u>Closing Documents</u>. The Purchaser shall have delivered or caused to be delivered to the Seller a copy of the resolutions duly and validly adopted by the Board of Directors of the Purchaser (or a duly authorized committee thereof), certified by its Secretary, authorizing and approving the execution, delivery and performance of this Agreement and the transactions contemplated hereby.

8.3 Termination.

(a) This Agreement may be terminated at any time prior to Closing: (i) by mutual written agreement of the Seller and the Purchaser; (ii) by either the Seller or the Purchaser if Closing shall not have been consummated on or before December 31, 2004, for any reason other than delay in or nonperformance of the covenants or obligations of the party seeking termination, provided, however, that the foregoing date may be extended to January 31, 2005 in

the reasonable discretion of the Seller; or (iii) by either the Seller or the Purchaser if consummation of the transactions contemplated hereby would violate any nonappealable final order, decree or judgment of any Court or Governmental Authority having competent jurisdiction. The party desiring to terminate this Agreement pursuant to Sections 8.3(a)(ii) or 8.3(a)(iii) shall provide notice of such termination to the other party.

- (b) If the Agreement is terminated as permitted by Section 8.3(a)(i), such termination shall be without liability of any party (or any stockholder, director, officer, employee, agent, consultant or representative of such party) to any other party to this Agreement; provided, however, that if such termination shall result from the willful (i) failure of either party to fulfill a condition to the performance of the obligations of the other party, (ii) failure to perform a covenant of this Agreement or (iii) breach by either party hereto of any representation or warranty or agreement contained herein, such party shall be fully liable for any and all Damages incurred or suffered by the other party as a result of such failure or breach. The provisions of Articles IX and X shall survive any termination hereof pursuant to Section 8.3(a).
- Money shall not be repaid by the Seller to the Purchaser; provided, however, that the Seller shall repay the Earnest Money to the Purchaser in the event that the Agreement is terminated under (i) Section 8.3(a)(ii) solely due to failure of the conditions described in Sections 8.1(a), 8.1(b), 8.1(d)(ii), 8.1(d)(ii), 8.1(d)(iv) or 8.1(f) to be satisfied or the failure of the condition described in Section 8.2(b) to be satisfied or (ii) Section 8.3(a)(iii). Without limiting the foregoing, in the event that this Agreement shall be terminated under Section 8.3(a)(ii) solely due to the failure of the condition described in Section 8.1(e) to be satisfied, the Seller shall retain the Earnest Money as the sole and exclusive remedy for Damages with respect to termination of this Agreement.

ARTICLE IX

INDEMNIFICATION

9.1 <u>Survival</u>. The covenants, agreements, representations and warranties of the parties contained in this Agreement or in any certificate or other writing delivered pursuant hereto or in connection herewith shall survive the Closing until the date which is 18 months after the Closing Date (the "Survival Date") (other than those covenants and agreements that by their terms survive the Survival Date, in which case such covenant or agreement shall survive until the applicable date thereof). Notwithstanding the preceding sentence, any covenant, agreement, representation or warranty in respect of which indemnity may be sought under this Agreement shall survive the time at which it would otherwise terminate pursuant to the preceding sentence, if notice of the inaccuracy or breach thereof giving rise to such right of indemnity shall have been given to the party against whom such indemnity may be sought prior to such time.

9.2 Indemnification.

(a) The Seller hereby indemnifies the Purchaser against and agrees to hold it harmless from any and all damage, loss, liability and expense (including, without limitation, reasonable expenses of investigation and reasonable attorneys' fees and expenses actually

incurred in connection with any action, suit or proceeding whether involving a third party claim or a claim solely between the parties hereto) ("Damages") incurred or suffered by the Purchaser arising out of (i) any misrepresentation or breach of warranty or breach of covenant or agreement made or to be performed by the Seller pursuant to this Agreement, (ii) the Retained Assets or the Retained Liabilities or (iii) the ownership and operation of the Business prior to the Closing (except that the Seller shall not have any liability for any Damages arising out of or with respect to the condition (including breaches) or suitability for use of the Assets (other than pursuant to clause (i) of Section 9.2(a))); provided, however, that (A) the Purchaser shall not make a claim against the Seller for indemnification under Section 9.2 for Damages unless and until the aggregate amount of such Damages exceeds \$100,000 (the "Seller Basket"), in which event the Purchaser may claim indemnification only for Damages in excess of the Seller Basket and (B) the aggregate maximum liability of the Seller under clause (i) of this Section 9.2(a) shall not exceed \$6,875,000.

(b) The Purchaser hereby indemnifies the Seller against and agrees to hold the Seller harmless from any and all Damages incurred or suffered by the Seller arising out of (i) any misrepresentation or breach of warranty or breach of covenant or agreement made or to be performed by Purchaser pursuant to this Agreement and (ii) the ownership and operation of the Business after the Closing; provided, however, that (A) the Seller shall not make a claim against the Purchaser for indemnification under this 9.2(b) for Damages unless and until the aggregate amount of such Damages exceeds \$100,000 (the "Purchaser Basket"), in which event the Seller may claim indemnification for Damages in excess of the Purchaser Basket and (B) Purchaser's maximum liability under clause (i) of this Section 9.2(b) shall not exceed \$6,875,000.

9.3 Procedures.

- (a) The party seeking indemnification under Section 9.2 (the "Indemnified Party") agrees to give prompt notice to the party against whom indemnity is sought (the "Indemnifying Party") of the assertion of any claim, or the commencement of any suit, action or proceeding ("Claim") in respect of which indemnity may be sought and will provide the Indemnifying Party such information with respect thereto that the Indemnifying Party may reasonably request. The failure to so notify the Indemnifying Party shall not relieve the Indemnifying Party of its obligations hereunder, except to the extent such failure shall have adversely prejudiced the Indemnifying Party.
- (b) Subject to the limitations set forth in this Section 9.3, the Indemnifying Party shall be entitled to control the defense, and appoint lead counsel for such defense, of any Claim asserted by any third party ("Third Party Claim"), in all cases at the Indemnified Party's expense.
- (c) If the Indemnifying Party assumes the control of the defense of any Third Party Claim in accordance with the provisions of this Section 9.3, (i) the Indemnifying Party shall obtain the prior written consent of the Indemnified Party (which shall not be unreasonably withheld) before entering into any settlement of such Third Party Claim, if the settlement does not include an unconditional release of the Indemnified Party from all liabilities and obligations with respect to such Third Party Claim and (ii) the Indemnified Party shall be entitled to participate in the defense of such Third Party Claim and to employ separate counsel of its choice

for such purpose. The fees and expenses of such separate counsel shall be paid by the Indemnified Party.

- (d) Each party shall cooperate, and cause their respective Affiliates to cooperate, in the defense or prosecution of any Third Party Claim and shall furnish or cause to be furnished such records, information and testimony, and attend such conferences, discovery proceedings, hearings, trials or appeals, as may be reasonably requested in connection therewith.
- (e) Each Indemnified Party shall use reasonable efforts to collect any amounts available under insurance coverage, or from any other Person alleged to be responsible, to mitigate any claim or indemnity for any Damages payable under Section 9.2.
- Notwithstanding any provision in this agreement to the contrary, the Seller shall have the sole right, with counsel of its choice, to defend and/or settle any claim, action or proceeding (i) in which a Governmental Authority has instituted or asserted a claim, whether directly or indirectly (including, without limitation, by counterclaim, cross-claim, third party claim, interpleader or otherwise), (ii) seeking injunctive relief, (iii) involving a class action, (iv) involving allegations of criminal activities, or (v) involving allegations of violations of the Racketeering Influenced and Corrupt Organizations Act, as amended or any securities or antitrust Law (a "Non-Assumable Claim") and Purchaser will not be entitled to assume defense thereof; provided, however, that the Seller shall consult with Purchaser before settling any Non-Assumable Claim. If the Seller assumes the defense of a Non-Assumable Claim, it must pursue such defense, settlement or negotiation diligently and in good faith. Upon the receipt by the Seller of an offer of compromise relating to such Non-Assumable Claim that includes an unconditional release of the Seller from, or upon entry of a final judgment with respect to, such Non-Assumable Claim (and such offer to compromise or judgment requires primarily the payment of money), the Seller shall promptly inform Purchaser of such offer or judgment, together with a description of the material terms and conditions of such offer or judgment. Purchaser shall have the right to terminate its liability for Damages with respect to the Non-Assumable Claim that is the subject of such an unconditional offer of compromise or judgment upon its irrevocable offer to the Seller to pay the amount contained in such offer to compromise or judgment. Upon receipt, in the form of immediately available funds, by the Seller of the amount contained in such offer to compromise or judgment and payment of all other Damages suffered or incurred by it in connection with such Non-Assumable Claim, Purchaser shall have no further liability to the Seller with respect to such Non-Assumable Claim.

9.4 <u>Calculation of Damages.</u>

(a) The amount of any Damages payable under Section 9.2 by the Indemnifying Party shall be net of any (i) amounts recovered or recoverable by the Indemnified Party under applicable insurance policies or from any other Person alleged to be responsible therefor and (ii) any current tax benefit realized therefrom. If the Indemnified Party receives any amounts under applicable insurance policies, or from any other Person alleged to be responsible for any Damages, subsequent to an indemnification payment by the Indemnifying Party, then such Indemnified Party shall promptly reimburse the Indemnifying Party for any payment made or expense incurred by such Indemnifying Party in connection with providing such

indemnification payment up to the amount received by the Indemnified Party, net of any expenses incurred by such Indemnified Party in collecting such amount.

- (b) Notwithstanding any other provision of this Agreement to the contrary, if at the Closing the Indemnified Party knows or has been advised by the Indemnifying Party that one or more of the representations and warranties made by the Indemnifying Party is inaccurate as of the Closing Date or the date made, then the Indemnified Party shall have no right or remedy after the Effective Time with respect to such inaccuracy and shall be deemed to have waived its rights to indemnification in respect thereof.
- 9.5 <u>Assignment of Claims</u>. Without limiting the generality of the last sentence of Section 9.4, if the Indemnified Party receives any payment from an Indemnifying Party in respect of any Damages pursuant to Section 9.3 and the Indemnified Party could have recovered all or a part of such Damages from a third party (a "*Potential Contributor*") based on the underlying Claim asserted against the Indemnifying Party, the Indemnified Party shall assign such of its rights to proceed against the Potential Contributor as are necessary to permit the Indemnifying Party to recover from the Potential Contributor the amount of such payment
- 9.6 <u>Fraud; Intentional Misrepresentation</u>. Notwithstanding any provision to the contrary contained in this Agreement, no limitation on liability shall apply to any claim based on fraud or intentional misrepresentation.
- 9.7 <u>Sole Remedy</u>. Except as otherwise expressly provided in this Agreement, the Seller and the Purchaser hereby acknowledge and agree that the sole and exclusive remedy of the Seller and the Purchaser with respect to any and all Damages relating to the subject matter of this Agreement shall be pursuant to the indemnification provisions set forth in this Article 9. Except as set forth in this Agreement, neither the Seller nor the Purchaser is making any representation, warranty, covenant or agreement with respect to the matters contained herein.

ARTICLE X

MISCELLANEOUS

Notices. All notices, requests, consents and other communications hereunder shall be in writing, shall be addressed to the receiving party's address set forth below or to such other address as a party may designate by notice hereunder, and shall be delivered by hand or recognized overnight courier.

If to the Seller: Advanced TelCom Group, Inc.

c/o GE Vendor Financial Services

10 Riverview Drive Danbury, CT 06810 Attn: General Counsel

30

Facsimile: (203) 749-4534

TRADE

And a copy to: King & Spalding LLP

1185 Avenue of the Americas New York, NY 10036-4003 Attn: Roth Kehoe, Esq. Facsimile: (212) 556-2222

If to the Purchaser: Eschelon Telecom, Inc.

730 2nd Ave. S. Suite 900

Minneapolis, MN 55402

Attn: President

Facsimile: (612) 436-6726

With a copy to: Piper Rudnick LLP

1775 Wiehle Avenue

Suite 400

Reston, Virginia

Attn: Edwin M. Martin, Esq. Facsimile: (703) 773-5000

All notices, requests, consents and other communications hereunder shall be deemed to have been delivered (a) if by hand, at the time of the delivery thereof to the receiving party at the address of such party set forth above or (b) if sent by overnight courier, on the next Business Day following the day such notice is delivered to the recognized overnight courier service.

- 10.2 Entire Agreement. This Agreement and the Confidentiality Agreement embody the entire agreement and understanding between the parties hereto with respect to the subject matter hereof and supersedes all prior oral or written agreements and understandings relating to the subject matter hereof. No statement, representation, warranty, covenant or agreement of any kind not expressly set forth in this Agreement or the Related Agreements shall affect, or be used to interpret, change or restrict, the express terms and provisions of this Agreement or the Related Agreements.
- 10.3 <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors, heirs, personal representatives, legal representatives, and permitted assigns.
- 10.4 <u>Assignment</u>. Neither this Agreement, nor any right hereunder, may be assigned by any of the parties hereto without the prior written consent of the other parties, except that the Purchaser may assign all or part of its rights and obligations under this Agreement and any of the Related Agreements to one or more direct or indirect Subsidiaries without consent of any party hereto or thereto.
- 10.5 <u>Modifications and Amendments</u>. The terms and provisions of this Agreement may be modified or amended only by written agreement executed by all parties hereto.

- 10.6 Waivers and Consents. The terms and provisions of this Agreement may be waived, or consent for the departure therefrom granted, only by written document executed by the party entitled to the benefits of such terms or provisions. No such waiver or consent shall be deemed to be or shall constitute a waiver or consent with respect to any other terms or provisions of this Agreement, whether or not similar. Each such waiver or consent shall be effective only in the specific instance and for the purpose for which it was given, and shall not constitute a continuing waiver or consent. No failure or delay by a party hereto in exercising any right, power or remedy under this Agreement, and no course of dealing between the parties hereto, shall operate as a waiver of any such right, power or remedy of the party. No single or partial exercise of any right, power or remedy under this Agreement by a party hereto, nor any abandonment or discontinuance of steps to enforce any such right, power or remedy, shall preclude such party from any other or further exercise thereof or the exercise of any other right, power or remedy hereunder. The election of any remedy by a party hereto shall not constitute a waiver of the right of such party to pursue other available remedies. No notice to or demand on a party not expressly required under this Agreement shall entitle the party receiving such notice or demand to any other or further notice or demand in similar or other circumstances or constitute a waiver of the rights of the party giving such notice or demand to any other or further action in any circumstances without such notice or demand.
- 10.7 No Third Party Beneficiary. Except with respect to Article VII and as provided in Section 9.1, nothing expressed or implied in this Agreement is intended, or shall be construed, to confer upon or give any Person other than the parties hereto and their respective heirs, personal representatives, legal representatives, successors and permitted assigns, any rights or remedies under or by reason of this Agreement.
- 10.8 Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law, or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner adverse to any party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties hereto shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the extent possible.
- 10.9 <u>Publicity</u>. No party to this Agreement shall make, or cause to be made, any press release or public announcement or filing in respect of this Agreement or the transactions contemplated hereby or otherwise communicate with any news media without the prior written consent of the Purchaser and the Seller, except as may be required by Law or any listing agreement related to the trading of the shares of such party on any national securities exchange or national automated quotation system, in which case the party proposing to issue such press release or make such public announcement or filing shall use reasonable efforts to consult in good faith with the other party before issuing any such press release or making such public announcement or filing.
- 10.10 Governing Law; Jurisdiction. This Agreement and the rights and obligations of the parties hereunder shall be construed in accordance with and governed by the internal laws

of the State of New York, without giving effect to the conflict of law principles thereof. Except as otherwise expressly provided in this Agreement, the parties hereto agree that any suit, action or proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby shall be brought in the United States District Court for the Southern District of New York so long as one of such courts shall have subject matter jurisdiction over such suit, action or proceeding, and that any cause of action arising out of this Agreement shall be deemed to have arisen from a transaction of business in the State of New York, and each of the parties hereby irrevocably consents to the jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such suit, action or proceeding and irrevocably waives, to the fullest extent permitted by law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or proceeding in any such court or that any such suit, action or proceeding which is brought in any such court has been brought in an inconvenient forum. Process in any such suit, action or proceeding may be served in any manner authorized by the laws of the State of New York. Judgment upon any award may be entered in any court having jurisdiction. EACH OF THE PARTIES HERETO HEREBY IRREVOCABLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT, THE RELATED AGREEMENTS OR THE TRANSACTIONS CONTEMPLATED THEREBY.

- 10.11 Force Majeure. In the event that the Seller or the Purchaser is prevented from performing or is unable to perform any of its obligations under this Agreement due to any Act of God, fire, casualty, flood, earthquake, war, strike, lockout, epidemic, destruction of production facilities, riot, insurrection, act of terrorism, material unavailability, or any other cause beyond the reasonable control of the party invoking this Section 10.11, and if such party shall have used commercially reasonable efforts to mitigate its effects, such party shall give prompt written notice to the other party, its performance shall be excused, and the time for the performance shall be extended for the period of delay or inability to perform due to such occurrences. Notwithstanding the foregoing, if such party is not able to perform within 60 calendar days after the event giving rise to the excuse of Force Majeure, the other party may terminate this Agreement.
- 10.12 <u>Execution by Facsimile; Counterparts</u>. This Agreement may be executed by facsimile and in any number of counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same agreement.
- 10.13 <u>Headings and Captions</u>. The headings and captions of the various subdivisions of this Agreement are for convenience of reference only and shall in no way modify or affect the meaning or construction of any of the terms or provisions hereof.
- 10.14 Expenses. Except as otherwise specified in this Agreement, all costs and expenses, including fees and disbursements of counsel, financial advisors and accountants, incurred in connection with this Agreement and the transactions contemplated hereby shall be paid by the party incurring such costs and expenses, whether or not the Closing shall have occurred. Notwithstanding any provision to the contrary contained herein, (i) the Seller, on the one hand, and the Purchaser, on the other hand, shall each pay one half of all fees and expenses of Kelley, Drye & Warren LLP and (ii) the Purchaser shall pay all costs and expenses (including attorneys fees and expenses), or reimburse the Seller if Seller or ATI pays subject to prior

authorization from the Purchaser which shall not be unreasonably withheld, with respect to the Audit and with respect to obtaining any required authorizations, consents, orders, approvals, guaranties, bonds and estoppel certificates, including the authorizations, consents, orders, approvals, guaranties, bonds, letters of credit and estoppel certificates described on <u>Schedule</u> 8.2(b).

10.15 <u>Interpretation</u>. The parties hereto acknowledge and agree that (i) each party and its counsel, if so represented, reviewed and negotiated the terms and provisions of this Agreement, excluding Schedules, and have contributed to its revision and (ii) the rule of construction to the effect that any ambiguities are resolved against the drafting party shall not be employed in the interpretation of this Agreement.

10.16 Further Assurances. From time to time after the Closing Date, at the reasonable request of the other party to this Agreement and at the expense of the party so requesting, each of the parties to this Agreement shall execute and deliver to such requesting party such documents and take such other action as such requesting party may reasonably request in order to consummate more effectively the transactions contemplated by this Agreement. In addition, after the Closing Date, (a) any amounts received by the Seller or any of its Affiliates with respect to operation of the Business after the Closing shall be held by the Seller or such Affiliate for the benefit of the Purchaser and delivered within ten business days after receipt to the Purchaser and (b) any amounts received by the Purchaser or any of its Affiliates (including ATI and SCS) after the Closing with respect to operation of the Business prior to the Closing (other than with respect to Assets, including any amounts received pursuant to Accounts Receivable) shall be held by the Purchaser or such Affiliate for the benefit of the Seller and delivered within ten business days of receipt to the Seller.

[signatures on next page]

34

IN WITNESS WHEREOF, the parties hereto have each executed and delivered this Agreement as of the day and year first above written.

ADVANCED TELCOM GROUP, INC.

David M. O'Neill

President

ESCHELON TELECOM, INC.

By_____

Richard A. Smith President and CEO

IN WITNESS WHEREOF, the parties hereto have each executed and delivered this Agreement as of the day and year first above written.

ADVANCED TELCOM GROUP, INC.

By_

David M. O'Neill

President

ESCHELON TELECOM, INC.

Richard A. Smith

President and CEO

Exhibit A

GUARANTY

GUARANTY (the "Guaranty"), dated as of December ____, 2004 of GENERAL ELECTRIC CAPITAL CORPORATION ("GE Capital") in favor of ESCHELON TELECOM, INC. ("Eschelon").

WHEREAS, Eschelon and Advanced Telecom Group, Inc. ("ATG") have entered into that certain Stock Purchase Agreement dated as of October 13, 2004 (the "Purchase Agreement"); and

WHEREAS, Eschelon has requested GE Capital, as parent of ATG, to provide a guaranty to Eschelon on the terms and conditions hereinafter provided; and

WHEREAS, GE Capital is willing to enter into this Guaranty to induce Eschelon to enter into the Purchase Agreement;

NOW, THEREFORE, GE Capital hereby agrees:

Section 1. Guaranty by GE Capital. (a) From and after the date hereof, GE Capital hereby guarantees the full and prompt payment, performance and satisfaction by ATG of all of ATG's liabilities and obligations to Eschelon under the Purchase Agreement, in each case after any applicable grace periods or notice requirements in accordance with the terms of the Purchase Agreement (collectively, the "GE Obligations"); provided, however, that no performance or payment hereunder shall be required until at least fifteen (15) Business Days (as used herein, a "Business Day" shall refer to a day other than a Saturday or a Sunday on which commercial banks are open for business in New York City) after written notice from Eschelon to GE Capital indicating the specific obligations to be fulfilled by GE Capital hereunder. GE Capital hereby agrees that it will ensure that the GE Obligations will be satisfied strictly in accordance with their terms. The obligations of GE Capital hereunder shall be absolute and unconditional irrespective of the absence of any action to enforce the same against ATG or any other person or entity or any other event or circumstance that may otherwise constitute a legal or equitable discharge or defense of a guarantor. No amendment, waiver, consent, extension, forbearance or granting of any indulgence by Eschelon with respect to any provision of the Purchase Agreement shall affect the enforceability of GE Capital's guaranty of the GE Obligations, provided that GE Capital's guaranty hereunder shall be a guaranty of full and prompt payment, performance and satisfaction of the GE Obligations after giving effect to such amendment, waiver, consent, extension, forbearance or granting of any indulgence by Eschelon. GE Capital covenants that this Guaranty will not be discharged except by complete performance of the obligations contained in this Guaranty.

(b) GE Capital shall be subrogated to all rights of Eschelon in respect of any amounts paid by GE Capital pursuant to the provisions of this Guaranty; provided, however, that GE Capital shall be entitled to enforce, or to receive any payments arising out of or based upon, such

right of subrogation only after all of the GE Obligations have been paid, performed or satisfied in full.

- (c) This Guaranty shall continue to be effective or be reinstated, as the case may be, if at any time any payment of any GE Obligation is rescinded or must otherwise be returned by Eschelon upon the insolvency, bankruptcy or reorganization of GE Capital, ATG or otherwise, all as though such payment had not been made.
- Section 2.1. <u>Notices</u>. All notices to GE Capital under this Guaranty and copies of all notices to ATG under the Purchase Agreement shall, until GE Capital furnishes written notice to the contrary, be in writing and mailed, faxed or delivered to GE Capital at 260 Long Ridge Road, Stamford, Connecticut 06927-9400, and directed to the attention of the Senior Vice President-Corporate Treasury and Global Funding Operation of GE Capital (facsimile no. (203) 357-4975).
- Section 2.2. Governing Law. This Guaranty shall be construed and enforced in accordance with, and governed by, the laws of the State of New York, United States of America.
- Section 2.3. <u>Interpretation.</u> The headings of the sections and other subdivisions of this Guaranty are inserted for convenience only and shall not be deemed to constitute a part hereof.
- Section 2.4. <u>Attorney's Cost</u>. GE Capital agrees to pay all reasonable attorney's fees and disbursements and all other reasonable and actual costs and expenses which may be incurred by Eschelon in the enforcement of this Guaranty.
- Section 2.5. No Set-off. By acceptance of this Guaranty, Eschelon shall be deemed to have waived any right to set-off, combine, consolidate or otherwise appropriate and apply (i) any assets of GE Capital at any time held by Eschelon or (ii) any indebtedness or other liabilities at any time owing by Eschelon to GE Capital against, or on account of, any obligations or liabilities owed by GE Capital to Eschelon under this Guaranty.
- Section 2.6. <u>Currency of Payment</u>. Any payment to be made by GE Capital shall be made in US Dollars and such designation of the currency of payment is of the essence.

GENERAL ELECTRIC CAPITAL CORPORATION

By:_	
	Senior Vice President - Corporate Treasury
	and Global Funding Operation

ACKNOWLEDGMENT AND AGREEMENT

Eschelon Telecom, Inc. hereby acknowledges and consents to the provisions of the foregoing Guaranty.

RECORDED: 01/10/2007

ES	CHELON TELECOM, INC.	
Ву:	:Title:	-